Author's response to reviews

Title: Laparoscopic versus conventional appendectomy - a meta-analysis of randomized controlled trials

Authors:

Xiaohang Li (med328@163.com)
Jialin Zhang (jiz2200@126.com)
Lixuan Sang (sangliuxuan2008@sohu.com)
Wenliang Zhang (doctor512@sohu.com)
Zhiqiang Chu (chuzhiq@yahoo.com.cn)
Xin Li (pudding880@163.com)
Yongfeng Liu (yfliucmu@hotmail.com)

Version: 6 Date: 2 November 2010

Author's response to reviews: see over
Dear editor,

The description of the only one change made in the newest manuscript is listed below:

Reference is 17 instead of 7 on the third line in the Search Results section, on the seventh page.

There are no other errors in my manuscript after carefully checking.

Thank you very much.

Best wishes.

Yours Sincerely,

Xiaohang Li